Author’s response to reviews

Title: The prevalence of nutritional anemia in pregnancy in an east Anatolian province, Turkey

Authors:

leyla karaoglu (leyla.karaoglu@rize.edu.tr)
erkan pehlivan (epehlivan@inonu.edu.tr)
mucahit egri (megri@gop.edu.tr)
cihan deprem (cercan@inonu.edu.tr)
gulsen gunes (ggunes@inonu.edu.tr)
metin fikret genc (mgenc@inonu.edu.tr)
ismail temel (itemel@inonu.edu.tr)

Version: 2 Date: 23 April 2010

Author’s response to reviews: see over
Dear Editor,

Please find the attached documents of revised manuscript and our reply to the comments under the "revision note" heading.

We altered the manuscript in the direction of the suggestions made by the referee to the best we can and I hope the manuscript is eligible for publication in "Public Health" in its present form. Please accept our thanks and extend our gratitude to your editorial board and to our experts for the evaluation of the manuscript. Please do not hesitate to contact me if you require further information and I'm looking forward to hearing from you soon.

Sincerely yours,

Leyla Karaoglu, MD
Associate Professor

Corresponding address:
Rize University, Medical Faculty, Public Health Department, 53100, Rize, TURKEY
Fax: +90 464 212 30 09,
E-mail: lkaraoglu@inonu.edu.tr